Author’s response to reviews

Title: What makes or mars the facility-based childbirth experience: Thematic analysis of women’s childbirth experiences in western Kenya

Authors:

Patience Afulani (pafulani@ucla.edu; Patience.Afulani@ucsf.edu)

Leah Kirumbi (lwkirumbi@yahoo.com)

Audrey Lyndon (Audrey.Lyndon@ucsf.edu)

Version: 1 Date: 14 Dec 2017

Author’s response to reviews:

Dear Dr. Miller,

Thank you for the opportunity to revise and resubmit our manuscript. We have carefully revised the manuscript in response to the comments and provided a point by point response to each comment. We have attached:

1) One clean copy of the manuscript

2) One copy where your changes are highlighted (track changes)

3) A separate, point by point response to the editor and referee comments.

Sincerely,

Patience Afulani